Author’s response to reviews

Title: Women's knowledge, attitudes and practices (KAP) relating to breast and cervical cancers in rural Zimbabwe: A cross sectional study in Mudzi District, Mashonaland East Province

Authors:
Lovemore Makurirofa (makurirofal@gmail.com)
Priscilla Mangwiro (mangwiopriscilla@gmail.com)
Victoria James (victoria@nedico.co.zw)
Amos Milanzi (amos@nedico.co.zw)
Junior Mavu (manager@cancer.co.zw)
Michael Nyamuranga (nyamurangam@gmail.com)
Sydney Kamtauni (kamtaunis2207@gmail.com)

Version: 2 Date: 20 Sep 2018

Author’s response to reviews:

BMC editorial team

All the editorial comments were incorporated into the revised version of the manuscript and the authors’ responses are detailed in the cover letter hereby submitted as supplementary material. The changes can be seen in the tracked changes document submitted as supplementary material again.